Author's response to reviews

Title: Symptom burden in community-dwelling older people with multimorbidity: a cross-sectional study

Authors:

Jeanette Eckerblad (jeanette.eckerblad@liu.se)
Kersti Theander (kersti.theander@kau.se)
Anne Ekdahl (Anne.Ekdahl@lio.se)
Mitra Unosson (mitra.unosson@liu.se)
Ann-Britt Wirehn (Ann-Britt.Wirehn@lio.se)
Anna Milberg (anna.milberg@liu.se)
Barbro Krevers (barbro.krevers@liu.se)
Tiny Jaarsma (tiny.jaarsma@liu.se)

Version: 5
Date: 5 December 2014

Author's response to reviews: see over
The handling editor, Dr Stefano Volpato would like to have some changes. We recommend that you copyedit the paper to improve the style of written English.

<table>
<thead>
<tr>
<th>Comment</th>
<th>Changes in the text</th>
</tr>
</thead>
<tbody>
<tr>
<td>The handling editor, Dr Stefano Volpato would like to have some changes. We recommend that you copyedit the paper to improve the style of written English.</td>
<td>Thank you for the opportunity to improve the style of written English. We have followed your recommendation and the manuscript has now been proofread by an English native-speaking professional, Anchor English - Proofreading Services. <a href="http://www.anchorenglish.com">http://www.anchorenglish.com</a></td>
</tr>
</tbody>
</table>

**Best Regards**

On behalf of professor Tiny Jaarsma and professor Mitra Unosson.

Jeanette Eckerblad  
Department of Social and Welfare Studies,  
Faculty of Health Sciences, Linkoping University,  
601 74 Norrkoping Sweden